

### Research Article

## The Effect of Green Intellectual Capital on Green Performance in the Spanish Wine Industry: A Structural Equation Modeling Approach

# Bartolomé Marco-Lajara (), Patrocinio Zaragoza-Sáez (), Javier Martínez-Falcó (), and Lorena Ruiz-Fernández ()

Management Department, University of Alicante, Alicante, Spain

Correspondence should be addressed to Javier Martínez-Falcó; javier.falco@ua.es

Received 17 May 2022; Revised 3 July 2022; Accepted 15 July 2022; Published 8 August 2022

Academic Editor: Zhi-Qiang Jiang

Copyright © 2022 Bartolomé Marco-Lajara et al. This is an open access article distributed under the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited.

Global environmental problems, such as global warming, pollution, or deforestation, are critical issues that require a rapid and common response. In this context, companies play a decisive role in achieving environmental objectives through the ecological knowledge they can store and manage. In this context, the present research focuses its interest on analyzing how the set of green intangibles possessed by organizations, i.e., Green Intellectual Capital (GIC), affects their Green Performance (GP). Specifically, the study shows how GP is influenced by GIC through the mediating role of the Green Innovation (GI) variable. Therefore, the research questions to be answered by this study are as follows: Does GIC influence environmental performance? Does GI mediate the GIC-GP relationship? What actions can companies take to improve their GP? There are several reasons that have led us to carry out this research. First, there is little empirical evidence of the relationships proposed in this study. Second, to the best of our knowledge, there is no previous research that has contextualized the relationships raised in the wine industry, thus representing an advance in the comprehension of the constructs studied. Third, GIC represents an incipient field of study that needs to be developed and established within the literature linked to Intellectual Capital (IC). In order to achieve the proposed objectives, data from a survey of 202 wineries in Spain were used and a quantitative approach was followed using Structural Equation Modeling (PLS-SEM). The results of the research indicate that there is a positive and significant relationship between GIC and GP. In addition, GI partially mediates the relationship between these two variables, playing a key role in the environmental management of wineries. The theoretical and practical contributions of the study improve the understanding of the relationships raised, being a pioneering study due to its contextualization in the wine industry, as well as providing a series of guidelines for both environmental managers and winemakers to improve their GP.

#### 1. Introduction

Global warming, pollution, and deforestation are the manifestations of the accelerated growth of the world economy experienced in the last century [1]. These global environmental problems are critical and, therefore, require rapid and common responses, which has led to an intense academic debate on the need to achieve environmental sustainability [2].

Traditionally, activities aimed at environmental protection have conflicted with those aimed at economic performance [3]. On the one hand, earlier research claimed that the sole mission of managers was to maximize shareholder wealth [4]. On the other hand, recent studies indicate that companies must assume social and environmental responsibilities beyond purely economic ones [5, 6]. However, despite the ongoing debate on whether companies should be green or not, the need to balance their economic needs with environmental ones has begun to be recognized. This can be reflected in the growing interest of scholars in the study of environmental topics, such as green human resource management [7], green supply chain [8], sustainable manufacturing [9], circular economy [10], green creativity [11], and organizational sustainability [12].

In this context, Green Innovation (GI) represents a win-win solution to the economic performance-environmental management conflict [13, 14]. Concern for the environment has created great opportunities for companies to improve the environmental characteristics of their products and processes in order to gain competitive advantages [15]. In fact, numerous studies demonstrate the positive relationship between the development of GIs and the improvement of the economic, social, and environmental performance of companies [16]. This type of innovation requires the integration of internal and external resources through the development of organizational capabilities and intangible resources [17]. Therefore, companies wishing to properly manage their GIs should prepare themselves by acquiring the necessary environmental capabilities and resources.

There are two main reasons why there is a growing interest in environmental management in organizations: (1) stringent environmental regulations [18]; and (2) increased customer awareness [19]. These factors require an added effort on the part of organizations to reduce the environmental impact of their products and processes by promoting GI such as packaging or eco-design [20], as well as to communicate the environmental practices developed within the organization in an appropriate way [21].

Despite the widespread tendency of organizations to develop sustainable business practices, it is still unclear what are the drivers of GI that, in turn, drive Green Performance (GP) [22, 23]. From the Natural Resource-based View (NRBV), intangibles such as knowledge, skills, and capabilities of the firm constitute essential elements for its environmental management and, as a consequence, for obtaining a competitive advantage, since, under the NRBV logic, competitive advantage is achieved from the sustainable activities of the organization. Therefore, intangible assets related to environmental protection can be considered antecedents of GI and GP. To accumulate and use intangible assets, companies can adopt different approaches framed under the three dimensions of Intellectual Capital (IC): human, structural, and relational capital [24]. However, the IC that incorporates sustainable concepts, i.e., Green Intellectual Capital (GIC), has emerged as an important field of study at present [25], being the construct currently used by scholars to understand the cause-effect relationships between such set of green intangible assets and other green variables.

The GIC brings together the research fields of IC and corporate sustainability, being understood as "the total stock of all kinds of intangible assets, knowledge, capabilities, and relationships, etc., on environmental protection or green innovation at the individual level and at the organizational level within a company" [18]. Therefore, GIC presents clear differences with IC, given that, while the IC approach understands sustainability as one of several intangible assets, GIC considers sustainability as the core of intangible assets, as well as the main driver for knowledge generation and management.

In this context, the present study aims to investigate the relationship between GIC-GP, as well as the mediating role of GI in this relationship. Three main reasons justify the need to carry out this study. First, there is little research that has provided empirical evidence on the relationships discussed above. Second, no previous research has been identified that addresses the relationships to be raised in the industry under analysis, i.e., the wine industry. Third, GIC is an incipient field of study that needs to be developed and established in the field of environmental management of organizations. To this end, as Chen [18] points out, there is a need for research on how GIC drives other variables, such as GP and GI. The research is of use for both academics and wine industry professionals, since, on the one hand, the research contributes to the academic literature by addressing the study of environmental intangibles and their relationship with GP, contextualizing the study in a sector where such relationships had not been previously addressed, and, on the other hand, the research is helpful for wine industry professionals by offering a series of recommendations for winemakers to improve their GIC and, as a consequence, their GP.

Thus, the present research aims to fill the existing gap in the literature through the study of GIC to advance the understanding of this construct, as well as its relationship with GP. In addition, GI is conceived as a mediating variable in this relationship. This study, therefore, aims to shed light on the ambiguity surrounding the GIC construct. It contributes to the existing debate in the environmental management literature, first, by analyzing the effect of GIC on GP in the wine industry and, second, by exploring the mediating role of the GI variable in clarifying this relationship. In particular, the research aims to answer the following two research questions: (1) Does GIC have a positive effect on GP? (2) Does GI mediate the GIC-GP relationship? (3) What actions can wineries take to improve their GP?

To answer the research questions, the article is structured as follows: First, after this introduction, Section 2 reviews the literature on the GIC effect on GP and GI, formulating, in turn, the hypotheses of the model proposed. Section 3 describes the methodology followed to achieve the research objectives, highlighting aspects related to data collection, measurements, and the statistical technique used. Section 4 examines the theoretical hypotheses put forward by discussing the results. Finally, Section 5 presents the most important theoretical and practical implications of the research, as well as the limitations and future lines of research.

#### 2. Literature Review and Hypothesis Development

The widespread trend of organizations towards the development of more environmentally friendly business practices, as well as towards the generation of intangible assets and sustainable capabilities, such as GIC or GI, can be understood under the premises of the NRBV. The intellectual background of this approach has its origin in the seminal article written by Hart [26], who proposed this new vision under the logic of the Resource-based View (RBV), postulating that the strategy and competitive advantage of organizations in the coming years would be based on capabilities that facilitate sustainable economic activity.

The NRBV represented a milestone in the environmental management literature by relating for the first time organizational intangibles, such as tacit knowledge or capabilities, to the environmental management of companies, offering, in addition, a new perspective focused on studying the interaction between the company and its natural environment, given that, under the principles of the theory formulated, the natural environment could act as an entry barrier for companies to generate sustainable competitive advantages [27]. Therefore, under this approach, environmental management is considered an essential element in business strategy.

As a consequence of the above, the role of intangible assets and capabilities has attracted the interest of several authors specialized in environmental management. Thus, while Russo and Fouts [28] stressed the importance of organizational routines and capabilities for developing organizational policies to prevent pollution, Aragón-Correa and Sharma [29] emphasized that the implementation of a successful environmental strategy depended, in large part, on the development of a series of organizational capabilities in which accumulated knowledge, the vision shared among the members of the organization and learning processes played a fundamental role. More recently, authors such as Yusoff et al. [30] and Yong et al. [31] have focused on analyzing the influence of existing green intangible assets in the organization on the development and sustainable performance of organizations.

IC is the theoretical construct in which organizational intangible assets aimed at economic, social, and environmental improvement best fit. This concept refers to the set of intangible resources (human, structural, and relational) that allow a company to obtain a sustainable competitive advantage over time [32], and its study has been developed under the Intellectual Capital-based View (ICBV) [33]. According to RBV, a firm's intangible resources are more likely to contribute to achieving and maintaining superior performance when they are combined or integrated [34]. However, the two main criticisms of such a view revolve around: (1) the lack of specificity and (2) the lack of a clear explanation for achieving competitive advantage [35, 36]. ICBV aims to overcome the limitations of RBV with regard to the measurement and evaluation of intangible assets by having a greater explanatory capacity to address the relationship between organizational intangibles and their performance/competitive advantage [37, 38].

Although IC has been widely studied under the ICBV, recently some authors have attempted to integrate such an approach into the environmental discourse through the GIC construct [25, 30, 31, 39–42]. ICBV enables organizations to implement stringent international regulations, comply with the growing environmental awareness of consumers, and create value for the organization [43]. For this reason, its role is fundamental to ensuring the success of the Sustainable Development Goals (SDGs) promoted by the United Nations [44].

Definitions of GIC are scarce in the environmental management literature. On the one hand, Chen [18], in his pioneering work, defined it as the total set of intangible assets, knowledge, and capabilities related to environmental protection or green innovation at the individual and organizational levels within an organization. Chen [45], on the other hand, defined it as the integration of green and environmental knowledge sources into the organization to enhance its competitive advantage. Similarly, Liu [46] conceived it as the sum of all organizational knowledge that enables the organization to improve environmental management to gain a competitive advantage. López-Gamero et al. [47], on the other hand, conceptualized it as "the total stock of all kinds of intangible assets, knowledge, capabilities, relationships, etc., on environmental protection at the individual and organizational levels of the firm". Along the same lines, Huang and Kung [43] stated that "green intellectual capital represents a company's intangible assets, including knowledge wisdom, capabilities, experience, and innovation in the field of environmental protection".

It should also be noted that, as suggested by Hart [26], the GIC consists of three dimensions: Green Human Capital (GHC), Green Structural Capital (GSC), and Green Relational Capital (GRC). On the one hand, GHC is defined as the set of knowledge, skills, abilities, capabilities, experiences, and commitments of employees about environmental protection that is embedded in employees and not in organizations [18], allowing an organization to recognize its intangible assets related to the environment and helping to implement green strategies in a given competitive environment. On the other hand, GSC is conceived as the set of organizational assets that show concern for environmental protection within the company [48]. Thus, Jardon and Dasilva [49] suggest that the organization's concern for environmental aspects is not modified only by human capital, since the support of organizational culture and systems is required to increase the level of environmental awareness in the organization. Finally, GRC refers to the set of intangible assets based on the existing relationships between the organization and suppliers, customers, network members, and partners to improve the environmental management of the company and thus achieve a competitive advantage. Therefore, GRC plays a key role in building strong and lasting relationships between the organization and its customers and suppliers.

In addition, GIC provides a better understanding of the background of GI. The ability to innovate is becoming increasingly important for companies in a dynamic global environment, with more and more companies willing to devote their resources to the development of GI [50]. The development of GI is a win-win solution for all stakeholders, overcoming the conflict between economic development and environmental protection. The "green" label is an incentive for continuous innovation, as it creates new market opportunities for companies to meet new consumer demands and, as a consequence, improve their performance. GI encompasses innovation in energy-saving technologies, pollution prevention, waste recycling, eco-friendly product design, and environmental management of companies [51].

There are two types of GI: green product innovation and green process innovation. On the one hand, green product innovation aims to modify product designs by using nontoxic compounds or biodegradable materials during the production process to reduce the impact of disposal on the environment and improve energy efficiency [52]. This includes improvements in the durability or recyclability of products, reduction of raw materials, selection of more environmentally appropriate raw materials, and elimination of hazardous substances [53]. For its part, green process innovation aims to reduce energy consumption during the production process or during the process that converts waste into a valuable item [54]. In particular, green process innovation includes reducing air emissions, reducing water consumption, improving resource and energy efficiency, and switching from fossil fuels to bioenergy [53]. Thus, companies that pioneer green innovation strategies can achieve and maintain various competitive advantages [55], gaining not only cost efficiency, but also economic profitability [16].

Therefore, both GIC and GI can lead to improved GP, understood as the positive consequences of green initiatives on the natural environment inside and outside the company [18]. The present research attempts to link both the NRBV and the ICBV to understand the relationships posed in the study between the variables GIC, GI, and GP. The following sections set out the hypotheses put forward in the research.

2.1. Green Intellectual Capital and Green Performance. Companies cannot ignore the increasing environmental concerns of customers and the pressures of international environmental regulations [25]. In this context, companies must consider the negative externalities generated by their activities on the environment through GIC management, which can not only reduce production waste and increase productivity but also increase the GP of the organization. This can be understood as the degree to which the activities carried out by the organization are environmentally friendly [56, 57].

GHC refers to the set of intangibles that are strongly rooted in employees' skills, being owned by the members of the organization and not by the employees. Therefore, this set of intangible assets represents a solid basis for obtaining a competitive advantage, given the difficulty for competitors to imitate these resources [58]. Thus, the GHC not only allows obtaining a sustainable competitive advantage, but also improves the GP of the organization, given that by acquiring greater environmental skills, the GHC will be higher and, as a consequence, the GP will also increase [59].

Thus, if the green knowledge stock of the workers, i.e., the GHC, is higher, the GP of the organization will also be higher [60]. The GHC represents a key piece, therefore, for the success of the GP of organizations, since the knowledge and skills of employees are essential to address the environmental challenges faced by the organization [61]. If a company develops a strong GHC, employees will have the necessary green knowledge to address environmental challenges. In fact, in the environmental management literature, a qualified GHC is considered to increase the organization's chances of gaining a sustainable advantage over time, since a more environmentally aware workforce will have better skills to address such organizational challenges and, therefore, contribute to a greater extent to the generation of sustained performance over time [62, 63].

Furthermore, the interaction between companies can significantly increase their ability to address their environmental challenges [64], as close and intense linkages between companies can be an effective means for them to collaborate in reducing the negative externalities generated by their activities. Frequent interaction between external partners can encourage the organization to exchange resources and capabilities, as well as establish stronger relationships [65], and can generate positive externalities that improve GP [61]. Therefore, the GRC can improve the GP by matching the environmental interests of the main stakeholders with those of the organization [66]. In addition, trust between companies and their stakeholders plays a fundamental role in the acquisition of green knowledge, which is subsequently translated into improved GP [67]. Likewise, it should be noted that although organizations may have environmental objectives among their priorities [68], they need to crystallize their environmental management through their corporate objectives, culture, strategies, and organizational structure [69]. The GSC is thus understood as an internal resource that can improve the achievement of environmental objectives [70, 71], by representing a set of green intangibles owned by the organization that allows it to improve its GP.

The GIC can improve GP in different ways. First, it allows for minimizing environmental costs. Second, it promotes green knowledge and awareness among employees. Third, it enables the company's adherence to the demanding standards and expectations present in governmental environmental regulations. According to Chuang and Huang [72], GP is rooted in the organization's ability to accumulate knowledge and resources related to environmental protection. Therefore, based on the above arguments, we put forward the following hypothesis:

H1. GIC has a positive effect on GP.

2.2. Green Intellectual Capital and Green Innovation. Numerous studies claim that a company's IC positively influences its capacity for innovation [73–76]. Similarly, companies that properly manage their GIC have several intangibles that favor GI, enabling the company to adapt to new challenges related to sustainable development [76, 77].

Human capital is an essential resource for developing organizational innovations since the knowledge possessed by employees represents the key intangible resource for developing innovations to compete in the current turbulent environment [78, 79]. Therefore, GI will be higher as the green knowledge stock of employees is higher [60]. GHC thus acts as a catalyst for GI, enabling organizations to stimulate their green product and process innovations [30]. This set of environmental intangibles provides a competitive advantage to the firm through improved eco-innovation capability [51]. GHC improves environmental practices at the firm level [25], so firms with higher GHC tend to adopt more environmental innovations. Such a set of environmental intangibles have become a necessity to achieve environmental success for organizations [80], as companies wishing to develop GI need employees with high environmental knowledge. Also, GHC can generate an organizational climate in which risk, failure, and uncertainty are tolerated, thus facilitating the generation of creative ideas to develop GI. Therefore, if companies want to generate new green products and processes, they will need to develop their GHC.

Close relationships between employees and institutions facilitate innovation processes due to the faster flow of knowledge and its better utilization [81, 82]. In fact, numerous researchers claim that the relational capital of organizations, which is deployed through the exchange of ideas and knowledge between them, has a positive effect on organizational innovations [83-85]. From an environmental point of view, the tacit knowledge of employees [86], their social relationships [71], as well as inter-organizational relationships [8], serve to develop new technologies, ideas, products, and/or processes focused on preventing pollution caused by organizations, which can materialize in environmental product and process innovations. Therefore, the formation of a collaborative network between organizations to achieve environmental objectives, crystallized through the GRC, can foster the development of GI [87]. Therefore, as Huang and Kung [43] point out, GRC improves a company's cooperation and engagement with its customers, suppliers, and other stakeholders on environmental sustainability issues, which can translate into improved GI.

Despite developing GHC and GRC, if an organization does not have management systems and an adequate environmental culture, GI will be impossible to achieve. The elements of GSC, such as organizational structure, organizational culture, databases, and internal capabilities, directly increase the efficiency of GI, leading to higher firm profitability [88]. Moreover, organizational structure, culture, and policies lead to the improvement of innovation [89], so the integration of environmental knowledge in the organization can favor the development of GI. Thus, a company with a strong environmental culture fosters the acquisition of new green knowledge by employees, which is subsequently translated into GI [90]. Similarly, when such knowledge for the protection of the environment is codified, it can become systematically disseminated within the organization, to subsequently be used to develop GI [91]. Based on the above arguments, we propose the following hypothesis.

H2. GIC has a positive effect on GI.

2.3. Green Innovation and Green Performance. GI can not only increase the financial and social performance of companies but can also reduce the negative environmental impact generated by their activity [92], given that it is associated with the environmental management objectives of the organization [51, 93, 94]. Thus, GI is an essential capability for the successful environmental management of

NRBV advocates that pollution prevention, product stewardship, and sustainable development are key environmental strategies to improve GP and, as a consequence, gain competitive advantages [26, 27]. Therefore, organizations that are pioneers in the development of GI can achieve sustainable competitive advantages over time, since this type of innovation promotes the responsible and efficient use of raw materials in the production process, resulting in lower organizational costs [95], as well as improving product differentiation, which translates into higher consumer demand [96]. In this sense, it should be noted that GI allows for improving the ecological image of companies and, as a consequence, their GP [97], since they can incorporate ecological concepts in the design and packaging of their products [51]. In fact, the academic literature summarizes these two advantages under two dimensions: competitive advantage and GP. On the one hand, several studies affirm that GI, both in product and process, favors the achievement of competitive advantages [5, 70, 76]. On the other hand, scholars of the subject claim the existence of a positive relationship between GI and GP [91, 98, 99].

The implementation by organizations of GIs benefits organizations by enabling increased cost savings, improved environmental efficiency, and improved productivity, which directly contributes to competitive advantage [100]. In addition, the adoption of GIs potentially reduces pollution, hazardous toxic waste, and the cost of hazardous waste, while competently addressing external environmental pressures from other stakeholders with respect to environmental regulations [13, 100].

GP represents a major concern among managers since its improvement can imply compliance with strict environmental regulations and improve public perception of the products and services offered by the organization [101]. GI thus makes it possible to respond to the environmental needs of different stakeholders and to improve waste optimization. This is why companies that invest large resources to develop GI can: (1) reduce waste from their production, (2) increase their productivity, (3) comply with current regulations around environmental protection, (4) avoid sanctions from government agencies, (5) meet the environmental needs of stakeholders and (6) improve the differentiation of the organization. Therefore, in practice, the application of such type of innovation has the potential to improve the GP of organizations, with several researchers have demonstrated improved performance in terms of advantage, competitive green image, and GP [10, 12, 51, 102]. Thus, green products and process innovation are positively related to GP [100, 103, 104]. Based on the above arguments, we put forward the following hypothesis:

H3. GI has a positive effect on GP.

2.4. GI as a Mediating Variable in the GIC-GP Relationship. GHC refers to the set of knowledge, skills, and abilities possessed by employees in order to improve the environmental management of the company [10], favoring a climate conducive to tolerating failure and risk involved in the development of GI [12]. Thus, companies use the green knowledge of employees to develop GI to improve their GP [8]. Likewise, companies operating under strict environmental legislation understand the importance of GHC, given that this set of intangibles based on employees' skills, knowledge and creativity enables compliance with environmental regulations and encourages the adoption of strategies based on GI that, in turn, derive higher GP [55]. Thus, GHC allows a company to recognize its intangible green assets, facilitating the development and implementation of GIs to improve GP [61].

GRC, on the other hand, fosters GI development by reducing transaction and information search costs. Thus, when companies promote GRC in their environmental operations, they can better develop GI to minimize environmental impact and attract environmentally conscious customers. GRC is therefore a key resource for achieving the company's environmental goals from its relationships with suppliers, customers, and institutions [105]. Organizations should strive to integrate the green knowledge generated from the GRC, as GI is partly derived from this knowledge [106]. Therefore, GRC can increasingly increase GP through GI.

The GSC encompasses the existing codified environmental knowledge within an organization. This stock of organizational knowledge can favor GI, since the production of new products and the development of new processes often involve the application of this knowledge [107], positively influencing the GP of the organization [108]. Organizational structures and policies can positively influence innovations aimed at preserving the environment, and such innovations, in turn, can positively influence the GP of firms [109]. The codification of the environmental knowledge generated in the organization influences the GP of the organization, increasing the chances of successfully developing GIs [61]. Therefore, GSC, understood as the codified green knowledge possessed by the firm positively influences the GP of organizations by facilitating GI. Recently, Wang and Juo [61] demonstrated the mediating effect of GI on the GIC-GP relationship. However, there is little academic literature that has addressed such a mediating effect and, therefore, there is no certainty of such a relationship. Therefore, one of the main objectives of this study is to examine whether GI acts as a mediating variable in the GIC-GP relationship. Based on the above arguments, we propose the following hypothesis:

H4. GI mediates the relationship between GIC and GP. Figure 1 shows the theoretical model proposed with the hypotheses to be tested.

#### 3. Methodology

The methodological section of this research is divided into three blocks: (1) sample and population, (2) research questionnaire, and (3) analysis technique. Each of these blocks is described in detail below.



FIGURE 1: Proposed theoretical model. Source: own elaboration.

3.1. Sample and Population. Our research is contextualized in the Spanish wine industry for three main reasons. First, the Spanish wine industry has been selected as the unit of analysis given its importance for the economic and social development of Spain. Thus, according to the latest data provided by the International Organization of Vine and Wine (OIV), Spain stands out for its preferential position in the global wine industry, being the country with the largest vineyard surface area in the world and the largest exporter in volume. Second, in recent years various studies have emphasized the importance of knowledge as a key strategic factor in the modern wine industry [110–112]. Thus, while in the classic industrial approach, the wine sector was oriented to obtaining high yields per hectare as a formula for generating higher income, today, wineries are characterized by the intensive use of knowledge to reduce yields, prevent pests, enhance the expression of local varieties or take care of winemaking and aging in detail. Therefore, we consider it necessary to analyze green intangibles and their relationship with other constructs, being an increasingly knowledgeintensive industry. Third, no previous research has analyzed the relationships of the theoretical model proposed in the wine industry, which represents an opportunity to advance knowledge and understanding of the constructs under study.

The study population, therefore, is made up of companies in the Spanish wine industry, specifically those included in the National Code of Economic Activities (CNAE, for its Spanish acronym) 1102. According to the Sistema de Análisis de Balances Ibéricos (SABI) database, this CNAE code has a total of 4,373 companies, more than 99% of which are micro-enterprises and SMEs. A structured questionnaire based on the literature review was used for data collection in order to achieve greater representativeness in the results. First, the content of the questionnaire was validated through a pretest in which environmental managers of wineries and winemakers participated to validate the clarity and validity of the items used. Subsequently, the survey was distributed online using the Qualtrics application during the last 4 months of 2021. After the data collection process, a total of 216 observations were initially obtained. However, after the data cleaning process, 202 questionnaires were considered valid, this being the final sample (94.3% of the total number of responses). This sample size is valid for obtaining robust data through structural equation modeling, since a minimum sample size of 100 is required for the relationships between variables to be analyzed [113]. Likewise, it can be seen that the Spanish autonomous communities most represented in our sample are Castile and Leon (14.36%), Catalonia (13.37%), Castile La-Mancha (10.89%), and La Rioja (9.90%), all communities being represented to a greater or lesser extent in our sample (see Figure 2). This order is in line with the population under study, given that, according to the data provided by the SABI database, the communities with the most wineries are Castilla and León (17.35%), Catalonia (15.64%), and Castilla La-Mancha (11.31%). Furthermore, with regard to the size of the companies, 99.56% of the wineries in our sample are Small and Medium-sized Enterprises (SMEs), coinciding with the size of the companies belonging to the population studied, given that, based on the data on workers provided by the SABI database, 98.76% of Spanish wineries are SMEs.

3.2. Research Questionnaire. To ensure consistency, reliability, and validity in the measurement of the variables, scales previously validated online in previous research were used (see Table 1). For the measurement of GIC, we used the scale employed by Zaragoza-Sáez et al. [96], which was measured by the authors using a 7-point Likert-type scale with seven items, taking as a reference the measures provided in the studies conducted by Chen [18], Huang and Kung [43], and Chang and Chen [48]. For the measurement of GI, the scale used by Chen [45] was used, conceiving the construct as the development of environmentally friendly products and processes that modify the design of an existing product and/or process to reduce any negative impact on the environment. On the one hand, green product innovation refers to the introduction of environmentally friendly materials, environmentally friendly packaging, product recovery and recycling, and eco-labeling [51]. On the other hand, green process innovation relates to a company's ability to improve existing processes, as well as to develop new ones that generate savings and avoid pollution, save energy, favor waste recycling, or decrease toxicity in processes [51]. In that sense, we followed Chen's approach [45] by considering GI as a secondorder construct formed by green product innovation (4 items) and green process innovation (4 items) as first-order constructs. As with the GIC, the measurement scale used for this variable was Likert-type with seven response options (1–7). To measure GP, the 7-point Likert-type scale with 5 items by Paillé et al. [114] was used, which is based, in turn, on the five-item scale developed by Chow and Chen [115]. Finally, the size was introduced as a control variable, reporting the size of each organization according to the number of employees through the commonly known categorical classification that groups the different companies into microenterprises if they have up to 10 workers, small companies when they have 10 to 50 workers, medium-sized companies if they employ between 50 and 250 workers, and large companies when they have more than 250 workers [116].



FIGURE 2: Distribution of the companies in the sample by the autonomous community. Source: own elaboration.

3.3. Analysis Technique. The analysis technique used was partial least squares structural equation modeling (PLS-SEM), using the SmartPLS v. 3.3.3 software. This methodology makes it possible to analyze a network of theoretical relationships between variables, some of which may be latent, i.e., not observable. This makes PLS-SEM particularly useful for developing research in the field of social sciences since most of the concepts studied by the discipline are not directly observable [117]. In fact, the technique has experienced accelerated growth and acceptance in the field of social sciences during the last decade [118], particularly in the area of Management [119], the field in which our research is framed.

The choice of this technique for this study is based on several reasons. First, the minimum sample size for applying PLS-SEM is not very demanding [120], which is an advantage for the conduct of our study since the research sample (n = 202) is not very large. Thus, Reinartz et al. [121] consider that 100 observations should be the minimum sample size to give robustness to the results obtained in PLS-SEM. Second, our research establishes direct and indirect relationships between constructs, recommending the use of PLS-SEM in these cases as it can effectively handle these aspects [122]. Third, in our study, GI is a second-order construct, and this multidimensional variable can be effectively estimated in PLS [123]. Fourth, the PLS-SEM technique has been used previously to analyze the wine industry, so it is a valid research method for our investigation.

Likewise, it is important to note that there are several reasons to justify the use of the PLS-SEM technique as opposed to covariance-based structural equation modeling (CB-SEM). First, while CB-SEM is used when testing existing theory, PLS-SEM is appropriate in the exploratory phase for theory building and prediction [118]. Research addressing GIC, as well as its linkage to other constructs, is recent in the academic literature, so the field of study is still in the process of construction and consolidation. This justifies the preferential use of PLS-SEM over the CB-SEM approach in this study. Second, while for the use of the CB-SEM technique the properties of the normal distribution

Construct	Items	Measure	Source
Green intellectual	<ul> <li>GIC 1. Our employees care about the environment</li> <li>GIC 2. Our employees have the knowledge and skills to protect the environment</li> <li>GIC 3. Our employees cooperate in working groups to address environmental issues</li> <li>GIC 4. Our employees cooperate with our suppliers to protect the environment</li> <li>GIC 5. Our employees cooperate with our customers/distributors to protect the environment</li> <li>GIC 6. Our company implements innovations to protect the environment</li> <li>GIC 7. Our company invests in facilities to protect the environment</li> </ul>	Likert scale (1 = strongly	Zaragoza-sáez
capital (GIC)		disagree; 7 = strongly agree)	et al. (2020)
Green product innovation (GPTI) Green process innovation (GPSI)	<ul> <li>GPTI 1. The company chooses the product materials that produce the least amount of contamination to carry out product development or design</li> <li>GPTI 2. The company chooses the product materials that consume the least energy and resources to carry out product development or design</li> <li>GPTI 3. The company uses the least amount of materials to compose the product to carry out the development or design of the product</li> <li>GPTI 4. The company would deliberate with circumspection whether the product is easy to recycle, reuse and decompose to carry out product development or design</li> <li>GPSI 1. The emission of hazardous substances or wastes is effectively reduced in the manufacturing process</li> <li>GPSI 2. In the manufacturing process, waste and emissions are recycled to enable their treatment and reuse</li> <li>GPSI 3. Water, electricity, coal, or oil consumption is reduced in the manufacturing process</li> </ul>	Likert scale (1 = strongly disagree; 7 = strongly agree)	Chen (2008)
Green performance	<ul> <li>GP 1. Our company has reduced waste and emissions from operations compared to its competitors over the past 5 years</li> <li>GP 2. Our company has reduced the environmental impact of its products/services compared to its competitors over the last 5 years</li> <li>GP 3. Our company has reduced its environmental impact by establishing partnerships compared to its competitors over the past 5 years</li> <li>GP 4. Our company has reduced the risk of environmental accidents, spills, and emissions compared to its competitors over the last 5 years</li> <li>GP 5. Our company has reduced purchases of nonrenewable materials, chemicals, and components compared to its competitors over the past 5 years</li> </ul>	Likert scale (1 = strongly	Paillé et al.
(GP)		disagree; 7 = strongly agree)	(2014)

TABLE 1: Measurement of the variables analyzed.

Source: own elaboration.

must be strictly complied with, for the PLS-SEM technique it is not necessary to assume a normal distribution of the data as it is a nonparametric method [119]. Therefore, as far as the properties of the data distribution are concerned, the PLS-SEM technique has greater flexibility compared to CB-SEM. Third, the PLS-SEM technique is preferable when there are second-order variables in the model to be tested [117]. Given the multidimensional nature of the GI variable, consisting of green process innovation and green product innovation, the PLS-SEM approach is more appropriate than CB-SEM. Fourth, the minimum sample size of the study could allow us to apply both the PLS-SEM and the CB-SEM approach, since while the minimum sample size to apply PLS-SEM is 100, to apply CB-SEM a minimum sample size of 200 is required [118]. In this regard, it has been decided to select the PLS-SEM approach since the sample size is clearly higher than the minimum required, which increases the precision of the model estimates. Fifth, several previous investigations addressing the relationships between GIC, GI, and GP constructs have used PLS-SEM, which confirms the suitability of the technique to test the relationships proposed.

#### 4. Results

Given the multidimensional nature of the GI variable, we applied the two-stage model based on "latent variable scores" [124] to obtain the results. First, the aggregate scores of the first-order constructs were calculated. Second, these aggregate scores were used as indicators of the second-order construct. Next, based on the recommendations of Hair et al. [118], we present the results of the model in three blocks: (1) the evaluation of the global model, (2) the evaluation of the measurement model, and (3) the evaluation of the structural model.

4.1. Evaluation of the Global Model. The proposed model presents a standardized root mean square residual (SRMSR) of 0.047 < 0.08 [124], which means that the model has a good global fit. In addition, it also meets the more stringent requirement of Carmines and Zeller [125], who considers that SRMS must be less than 0.05 for there to be an adequate global fit of the model.

Once the SRMR fit criterion was checked, it was verified whether this indicator together with the unweighted least squares discrepancy (d\_ULS) and the geodesic discrepancy d\_G was within the confidence range after bootstrapping. As can be seen in Table 2, all values are below HI95 and HI99. Therefore, the results imply that this model cannot be rejected [118].

Table 3 shows the mean, maximum and minimum values, as well as the standard deviation for each variable are analyzed. As can be seen, the minimum and maximum values of the constructs GIC, GI, and GP are 1 and 7 respectively. This is because these are the minimum and maximum values of the Likert scale used. Likewise, while the minimum size of the wineries is 1, referring to the number of workers, the maximum value of the size variable is 262, which is the maximum value of the number of workers that make up a winery among the companies in the sample. Similarly, there are 10 workers on average among the companies in the sample, with the average of the three remaining variables being around values close to 5. Of the four constructs analyzed, the GIC is the one with the greatest dispersion to the average (1.496), while the size is the one with the least dispersion (0.969).

4.2. Measurement Model. To analyze the quality of the measurement model, the following criteria set forth by Hair et al. [118] were followed: (1) and analyze the reliability of the individual indicators through their external loadings ( $\lambda$ ), (2) assess the internal consistency reliability through Cronbach's alpha and composite reliability, (3) check the convergent validity through the average variance extracted (AVE) and (4) to analyze the discriminant validity the Heterotrait-Monotrait criterion (HTMT).

First, as indicated in Table 4, all the indicators of the variables analyzed do meet the requirement of individual

TABLE 2: Overall model fit.

	Value	HI95	HI99
SRMR	0.047	0.052	0.063
d_ULS	0.641	0.718	0.844
d_G	0.765	0.816	0.921

Source: compiled by authors.

TABLE 3: Values of the mean, minimum value, maximum value, and standard deviation of the variables analyzed.

	Mean	Min	Max	Standard deviation
GIC	4.944	1	7	1.496
GI	5.127	1	7	1.231
GP	4.873	1	7	1.343
Size	10.242	1	262	0.969

Source: compiled by authors.

item reliability, since their loadings exceed the value of 0.707 [126]. It is, therefore, possible to state that the different indicators present sufficient levels of reliability at the individual level. Second, it is possible to state that all the constructs meet the reliability criterion of internal consistency since both Cronbach's alpha and composite reliability  $(\rho c)$  exceed the value of 0.8. Third, the constructs meet the requirement of convergent validity, since their AVE measures exceed the 0.5 level [127]. Therefore, each construct explains more than half of the variance of its indicators. Finally, Table 5 shows the discriminant validity test following the HTMT criterion. As can be seen, the values of the GIC, GI, and GP are clearly lower than 0.85 [128]. This means that each construct is unique and, therefore, captures phenomena not represented by other constructs in the model.

4.3. Structural Model. Once it has been confirmed that the measures of the constructs are reliable and valid, the next step is to address the assessment of the structural model. This assessment involves examining the predictive ability of the model and the relationships between constructs. Following the indications followed by Gilinsky et al. [129], to evaluate the structural model we analyzed the path coefficients, the R-Squared level, and the predictive relevance of Q2.

First, before evaluating these indicators, we examined the presence of problems related to collinearity in the structural model. This is due to the need to avoid the presence of multicollinearity between the antecedent variables of each of the endogenous constructs. According to Hair et al. [118], there are indications of collinearity when the variance inflation factor is greater than 5 (VIF > 5). Therefore, values greater than five of the endogenous constructs imply critical levels of collinearity. In this sense, the VIF values obtained in this work do not exceed the maximum value in any of the cases (see Table 6). In addition, there is no unobserved heterogeneity in the sample data.

Through Figure 3 it can be observed how the results regarding *R*-Squared and  $\beta$  are based on a bootstrap test with 5000 subsamples. The direct and indirect effects of GIC on GP of wineries have been tested, finding both positive and

Construct/items	Outer loadings	Rho (Pa)	Cronbach's alpha	Ave
Green intellectual capital (GIC)		0.883	0.883	0.589
GIC 1. Our employees care about the environment	0.759			
GIC 2. Our employees have the knowledge and skills to protect the environment	0.781			
GIC 3. Our employees cooperate in working groups to address environmental issues	0.719			
GIC 4. Our employees cooperate with our suppliers to protect the environment	0.833			
GIC 5. Our employees cooperate with our customers/distributors to protect the	0.86			
environment.	0.00			
GIC 6. Our company implements innovations to protect the environment	0.71			
GIC 7. Our company invests in facilities to protect the environment	0.724			
Green innovation (GI)		0.948	0.945	0.725
GPTI 1. The company chooses the product materials that produce the least amount of	0.000			
contamination to carry out product development or design	0.882			
GPTI 2. The company chooses the product materials that consume the least energy and	0.003			
resources to carry out product development or design	0.903			
GPTI 3. The company uses the least amount of materials to compose the product to carry	0.991			
out the development or design of the product	0.001			
GPTI 4. The company would deliberate with circumspection whether the product is easy to	0.885			
recycle, reuse and decompose to carry out product development or design	0.005			
GPSI 1. The emission of hazardous substances or wastes is effectively reduced in the	0.00			
manufacturing process	0.89			
GPSI 2. In the manufacturing process, waste and emissions are recycled to enable their	0.001			
treatment and reuse	0.801			
GPSI 3. Water, electricity, coal, or oil consumption is reduced in the manufacturing process	0.838			
GPSI 4. In the manufacturing process, the use of raw materials is reduced	0.713			
Green performance (GP)		0.936	0.928	0.78
GP 1. Our company has reduced waste and emissions from operations compared to its	0.000			
competitors over the past 5 years	0.906			
GP 2. Our company has reduced the environmental impact of its products/services	0.027			
compared to its competitors over the last 5 years	0.937			
GP 3. Our company has reduced its environmental impact by establishing partnerships	0.706			
compared to its competitors over the past 5 years	0.780			
GP 4. Our company has reduced the risk of environmental accidents, spills, and emissions	0.97			
compared to its competitors over the last 5 years	0.07			
GP 5. Our company has reduced purchases of nonrenewable materials, chemicals, and	0.907			
components compared to its competitors over the past 5 years	0.907			

TABLE 4: Measurement model: external loadings, construct reliability, and convergent validity.

Source: compiled by authors.

TABLE !	5:1	Measurement	model:	discriminant	validity.

	GIC	GI	GP	Size
GIC				
GI	0.725			
GP	0.486	0.553		
Size	0.170	0.107	0.224	
0	9 11 4			

Source: compiled by authors.

TABLE 6: Analysis of collinearity in the model through VIF values.

	GIC	GP	GI	Size
GIC		1.778	1.000	
GP				
GI		1.779		
Size		1.013		

Source: compiled by authors.

statistically significant effects (see Table 7). This implies that the GI partially mediates the relationship between the GIC and GP, given the direct (0.169) and indirect (0.261) effects are positive and significant, with a strong total effect of GIC on GP of 0.430 ( $p \le = 0.01$ ). Therefore, all four hypotheses are supported. Table 8 also shows the effect sizes  $(f^2)$ , i.e., the degree to which an exogenous construct contributes to explaining a given endogenous construct in terms of  $R^2$ [130]. In this case, the most representative  $f^2$  values correspond to the GIC for the GI construct (0.775) and GI for the GP construct (0.108). Finally, to analyze the quality of the model, Geisser's test  $(Q^2)$  was performed, which must have estimated values above 0 ( $Q^2 > 0$ ). As can be seen in Table 9, an average predictive relevance of the model was observed because the  $Q^2$  values are above 0.25 [111]. Regarding the control variables, the results show that the size of the wineries has a positive and significant impact on the GP.



FIGURE 3: Theoretical model with *R*-squared, path coefficients ( $\beta$ ), and significance. Source: own elaboration.

TABLE 7: Results of the structural model for the mediation model.

Direct effects	Path coefficient	<i>t</i> -value	<i>p</i> -value	95% BCCI	Hypothesis supported
GI -> GP	0.391	3.635	$\leq = 0.01^{**}$	[0.195; 0.551]	H3 supported
GIC ->GI	0.666	13.437	$\leq$ = 0.01 <sup>**</sup>	[0.583; 0.748]	H2 supported
GIC ->GP	0.169	1.737	$0.042^{*}$	[0.069; 0.231]	H1 supported
Indirect effects	Path coefficient	t-value	<i>p</i> -value	95% BCCI	Hypothesis supported
$GIC \rightarrow GI \rightarrow GP$	0.261	3.572	$\leq$ = 0.01**	[0.134; 0.379]	H4 supported

Notes: BCCI: bias corrected confidence intervals; \*p < 0.05; \*\*p < 0.001. Source: compiled by authors.

TABLE 8: Effect sizes  $(f^2)$  of the analyzed variables.

	GP	GI
GIC	0.036	0.775
GI	0.108	
Size	0.048	

Source: compiled by authors.

TABLE 9: Construct cross validated redundancy.

	SSO	SSE	$Q^2$ (=1-SSE/SSO)
GI	1616	1103.561	0.317
GIC	1414	1414	
GP	1010	771.32	0.256
Size	202	202	

Source: compiled by authors.

Therefore, it could be interesting to analyze the differences in environmental management between large winery groups and small wineries.

The results of this research suggest that the development of GIC in wineries can lead to higher GP. Moreover, these intangibles not only generate greater ecological performance but also favor the formation of key organizational capabilities, such as GI, thus strengthening the GIC-GP relationship. The positive and significant relationship between GIC on GIP is in line with the results obtained by Yusliza et al. [66] and Wang and Juo [61], which demonstrate such a relationship for Malaysian and Taiwanese manufacturing industries, respectively. In addition, GIC acts as a catalytic variable for GI. In this regard, Chen & Chang [20]

demonstrate a positive and significant relationship between GHC and GI performance. In contrast, Chang [131] states that GHC does not directly affect GI, but indirectly through green adaptive capacity. Chen et al. [51] assert through their research that GIC has an indirect impact on GI through GSC. Ali et al. [119] show that GHC and GSC increase significantly with GI adoption. Jirakraisiri et al. [40], on the other hand, show that all three GIC dimensions positively and significantly influence the adoption of GI. In the present study, we analyzed the GIC construct in a unified way to understand the relationship of environmental intangibles to CG holistically. On the other hand, the mediating effect of GI on the GIC-GP relationship has been sparsely explored in the academic literature. However, through the joint understanding of the NRBV and ICV approaches, the mediating role that the GI variable can play in such a relationship has been corroborated.

#### 5. Conclusions and Implications

The results presented in the present research are relevant for the academic community, as well as for companies and professionals in the wine industry, as they improve the knowledge about the relationship between GIC and GP in the wine industry.

Wineries are facing increasing pressures to improve their environmental sustainability [132], as the environment, the community, and the local economy can be negatively affected by their activity. The wine industry is facing several exogenous factors that threaten its survival, such as rising energy prices, water scarcity, increasing environmental awareness among stakeholders, and climate change [133]. These factors, together with the knowledge of winemakers and wineries, can drive the adoption of sustainable practices that subsequently culminate in product and process innovations to improve the GP of wineries [129]. For this reason, the analysis proposed in the present research becomes particularly relevant, since the high impact of GIC on wineries' GP has been empirically demonstrated. Moreover, as demonstrated in the research, this positive effect is partially mediated by GI. Therefore, wineries that promote their GHC, GSC, and GRC will be able to improve their GP, as well as their capacity to develop innovations aimed at protecting the environment.

There are several theoretical and practical implications derived from our research. Regarding the theoretical contributions, the results of the present research contribute to the environmental management and IC literature, providing practical evidence in the Spanish wine industry. In particular, the research empirically demonstrates the positive and significant effect of GIC development on wineries' GP, as well as the mediating effect of GI on this relationship. These results are consistent with recent empirical research such as that of Wang and Juo [61] contextualized in Taiwan's hightech sector. However, further analysis of the constructs and their relationships should be pursued, as the academic literature addressing these relationships is very sparse. Therefore, we encourage environmental management researchers to continue to delve deeper into these relationships in future publications. The practical implications, therefore, allow us to answer the first two research questions, given that (1) there is a positive and significant relationship between GIC on GP, and (2) GI partially mediates this relationship. It is important to note that, to the best of our knowledge, no previous research has contextualized the relationships raised in the wine industry, which represents an advance in the understanding of the constructs studied, as well as in the comprehension of the environmental management of wineries.

Regarding the practical contribution of the study, the results presented in this research can play a key role in the environmental management decisions of environmental managers and winery winemakers. In this regard, from the GHC point of view, winery employees can develop codes of good environmental practices, organize training and environmental awareness sessions, as well as attend seminars and workshops to improve their green knowledge. As for the GSC, wineries can develop circular economy programs, computer systems to measure carbon and water footprint, eco-efficient facilities, a brand linked to sustainability, certifications that endorse their environmental commitment, a flat organizational structure through which green knowledge flows, an organizational culture built on the pillars of sustainability, as well as constant investments in R&D&i. As far as the GRC is concerned, the link between wineries and their stakeholders should be fostered, since such relationships can improve their green knowledge of the companies and, consequently, their environmental management. These organizational practices make it possible to accumulate a series of green intangibles that have a positive impact on the wineries' GP. Likewise, GI such as the development of organic wines, the technological improvement of agricultural soils, the valorization of waste, as well as the control of damages and climatic risks, can enhance the GIC-GP relationship. The practical implications provide an answer to the third research question, given that through actions aimed at improving employees' green knowledge (GHC), codifying the organization's environmental knowledge (GSC) and fostering relationships between different stakeholders (GRC), companies can improve their GP. The research can therefore be useful to winemakers who are thinking of improving their GP and/or their GIs, given that the actions proposed to improve the GHC, GSC, and GRC can lead to the improvement of the GP through the GI.

Despite the important contributions made in the article, it is important to highlight the existence of certain limitations. First, the relevance of the topic makes it necessary to extend this analysis to other wine-producing countries. In this sense, the effect of the GIC on the performance of wineries at the international level would be of great interest, and comparisons could be made between New World and Old World wine-producing countries. Second, there is a limitation inherent to cross-sectional studies, since they do not allow us to examine relationships over an extended period of time. Specifically, it would be interesting to know the evolution and temporal trajectory of the study carried out. For this reason, it seems relevant to us as a future line of research to analyze the companies that participated in the survey through longitudinal analysis. This implies that these companies should be willing to participate in the coming years in order to be able to investigate their evolution in an increasingly competitive, technological, and international business context. In addition, we could further investigate the role that winery size can play in GIC and GP. In fact, as a future line of research, we propose to carry out a multigroup analysis in which the differences in the model proposed according to the size of the wineries (SMEs or large companies) can be seen.

#### **Data Availability**

The data used to support the findings of this study are available from the corresponding author upon request.

#### **Conflicts of Interest**

The authors declare no conflicts of interest.

#### References

- Y. Jin, C. Cheng, and H. Zeng, "Is evil rewarded with evil? The market penalty effect of corporate environmentally irresponsible events," *Business Strategy and the Environment*, vol. 29, no. 3, pp. 846–871, 2020.
- [2] L. Hsueh, "Opening up the firm: what explains participation and effort in voluntary carbon disclosure by global businesses? An analysis of internal firm factors and dynamics," *Business Strategy and the Environment*, vol. 28, no. 7, pp. 1302–1322, 2019.
- [3] A. Scheidel, L. Temper, F. Demaria, and J. Martínez-Alier, "Ecological distribution conflicts as forces for sustainability:

an overview and conceptual framework," *Sustainability Science*, vol. 13, no. 3, pp. 585–598, 2018.

- [4] M. Friedman, *Capitalism and Freedom*, University of Chicago Press, Chicago, 1962.
- [5] C. H. Chang, "The influence of corporate environmental ethics on competitive advantage: the mediation role of green innovation," *Journal of Business Ethics*, vol. 104, no. 3, pp. 361–370, 2011.
- [6] R. Stewart and M. Niero, "Circular economy in corporate sustainability strategies: a review of corporate sustainability reports in the fast-moving consumer goods sector," *Business Strategy and the Environment*, vol. 27, no. 7, pp. 1005–1022, 2018.
- [7] A. A. Zaid, A. A. Jaaron, and A. Talib Bon, "The impact of green human resource management and green supply chain management practices on sustainable performance: an empirical study," *Journal of Cleaner Production*, vol. 204, pp. 965–979, 2018.
- [8] S. Yildiz Çankaya and B. Sezen, "Effects of green supply chain management practices on sustainability performance," *Journal of Manufacturing Technology Management*, vol. 30, no. 1, pp. 98–121, 2019.
- [9] S. H. Abdul-Rashid, N. Sakundarini, R. A. Raja Ghazilla, and R. Thurasamy, "The impact of sustainable manufacturing practices on sustainability performance," *International Journal of Operations & Production Management*, vol. 37, no. 2, pp. 182–204, 2017.
- [10] U. Awan and R. Sroufe, "Sustainability in the circular economy: insights and dynamics of designing circular business models," *Applied Sciences*, vol. 12, no. 3, p. 1521, 2022.
- [11] T. A. Bhutto, R. Farooq, S. Talwar, U. Awan, and A. Dhir, "Green inclusive leadership and green creativity in the tourism and hospitality sector: serial mediation of green psychological climate and work engagement," *Journal of Sustainable Tourism*, vol. 29, no. 10, pp. 1716–1737, 2021.
- [12] U. Awan, A. Khattak, S. Rabbani, and A. Dhir, "Buyer-driven knowledge transfer activities to enhance organizational sustainability of suppliers," *Sustainability*, vol. 12, no. 7, p. 2993, 2020.
- [13] M. Porter and C. van der Linde, "Green and competitive: ending the stalemate," *Harvard Business Review*, vol. 73, no. 5, pp. 120–134, 1995.
- [14] S. K. Singh, M. D. Giudice, R. Chierici, and D. Graziano, "Green innovation and environmental performance: the role of green transformational leadership and green human resource management," *Technological Forecasting and Social Change*, vol. 150, Article ID 119762, 2020.
- [15] H. Zameer, Y. Wang, and H. Yasmeen, "Reinforcing green competitive advantage through green production, creativity and green brand image: implications for cleaner production in China," *Journal of Cleaner Production*, vol. 247, Article ID 119119, 2020.
- [16] S. Asadi, S. OmSalameh Pourhashemi, M. Nilashi et al., "Investigating influence of green innovation on sustainability performance: a case on Malaysian hotel industry," *Journal of Cleaner Production*, vol. 258, Article ID 120860, 2020.
- [17] T. Lampikoski, M. Westerlund, R. Rajala, and K. Möller, "Green innovation games: value-creation strategies for corporate sustainability," *California Management Review*, vol. 57, no. 1, pp. 88–116, 2014.
- [18] Y. S. Chen, "The positive effect of green intellectual capital on competitive advantages of firms," *Journal of Business Ethics*, vol. 77, no. 3, pp. 271–286, 2007.

- [19] C. J. Chen and J. W. Huang, "Strategic human resource practices and innovation performance - the mediating role of knowledge management capacity," *Journal of Business Research*, vol. 62, no. 1, pp. 104–114, 2009.
- [20] Y. S. Chen and C. H. Chang, "Utilize structural equation modeling (SEM) to explore the influence of corporate environmental ethics: the mediation effect of green human capital," *Quality and Quantity*, vol. 47, no. 1, pp. 79–95, 2013.
- [21] E. Claver-Cortés, M. Dolores López-Gamero, J. F. Molina-Azorín, and P. Del Carmen Zaragoza-Sáez, "Intellectual and environmental capital," *Journal of Intellectual Capital*, vol. 8, no. 1, pp. 171–182, 2007.
- [22] S. Sharma and J. Aragón-Correa, Corporate Environmental Strategy and Competitive Advantage, Edward Elgar, Cheltenham, U.K.; North Hampton, MA, 2005.
- [23] N. Soewarno, B. Tjahjadi, and F. Fithrianti, "Green innovation strategy and green innovation," *Management Decision*, vol. 57, no. 11, pp. 3061–3078, 2019.
- [24] T. Davenport and L. Prusak, Working Knowledge: How Organizations Manage what They Know, Harvard Business Press, Cambridge, MA, 1998.
- [25] J. Y. Yong, M. Y. Yusliza, T. Ramayah, and O. Fawehinmi, "Nexus between green intellectual capital and green human resource management," *Journal of Cleaner Production*, vol. 215, pp. 364–374, 2019.
- [26] S. L. Hart, "A natural-resource-based view of the firm," *Academy of Management Review*, vol. 20, no. 4, pp. 986–1014, 1995.
- [27] S. L. Hart and G. Dowell, "Invited editorial: a natural-resource-based view of the firm," *Journal of Management*, vol. 37, no. 5, pp. 1464–1479, 2011.
- [28] M. V. Russo and P. A. Fouts, "A resource-based perspective on corporate environmental performance and profitability," *Academy of Management Journal*, vol. 40, no. 3, pp. 534–559, 1997.
- [29] J. A. Aragón-Correa and S. Sharma, "A contingent resourcebased view of proactive corporate environmental strategy," *Academy of Management Review*, vol. 28, no. 1, p. 71, 2003.
- [30] Y. M. Yusoff, M. K. Omar, M. D. Kamarul Zaman, and S. Samad, "Do all elements of green intellectual capital contribute toward business sustainability? Evidence from the Malaysian context using the Partial Least Squares method," *Journal of Cleaner Production*, vol. 234, pp. 626–637, 2019.
- [31] J. Y. Yong, M. Y. Yusliza, C. J. C. Jabbour, and N. H. Ahmad, "Exploratory cases on the interplay between green human resource management and advanced green manufacturing in light of the Ability-Motivation-Opportunity theory," *The Journal of Management Development*, vol. 39, no. 1, pp. 31– 49, 2019.
- [32] G. Martín-de-Castro, M. Delgado-Verde, P. López-Sáez, and J. E. Navas-López, "Towards 'an intellectual capital-based view of the firm': origins and nature," *Journal of Business Ethics*, vol. 98, no. 4, pp. 649–662, 2010.
- [33] K. K. Reed, M. Lubatkin, and N. Srinivasan, "Proposing and testing an intellectual capital-based view of the firm," *Journal* of Management Studies, vol. 43, no. 4, pp. 867–893, 2006.
- [34] J. Barney, "Firm resources and sustained competitive advantage," *Journal of Management*, vol. 17, no. 1, pp. 99–120, 1991.
- [35] R. L. Priem and J. E. Butler, "Is the resource-based "view" a useful perspective for strategic management research?" *Academy of Management Review*, vol. 26, no. 1, pp. 22–40, 2001.

- [36] N. J. Foss and T. Knudsen, "The resource-based tangle: towards a sustainable explanation of competitive advantage," *Managerial and Decision Economics*, vol. 24, no. 4, pp. 291–307, 2003.
- [37] M. A. Youndt, M. Subramaniam, and S. A. Snell, "Intellectual capital profiles: an examination of investments and returns," *Journal of Management Studies*, vol. 41, no. 2, pp. 335–361, 2004.
- [38] T. Andreeva, T. Garanina, J. Sáenz, N. Aramburu, and A. Kianto, "Does country environment matter in the relationship between intellectual capital and innovation performance?" *Journal of Business Research*, vol. 136, pp. 263–273, 2021.
- [39] S. Y. Malik, Y. Cao, Y. H. Mughal, G. M. Kundi, M. H. Mughal, and T. Ramayah, "Pathways towards sustainability in organizations: empirical evidence on the role of green human resource management practices and green intellectual capital," *Sustainability*, vol. 12, no. 8, p. 3228, 2020.
- [40] J. Jirakraisiri, Y. F. Badir, and B. Frank, "Translating green strategic intent into green process innovation performance: the role of green intellectual capital," *Journal of Intellectual Capital*, vol. 22, no. 7, pp. 43–67, 2021.
- [41] V. T. Dang and J. Wang, "Building competitive advantage for hospitality companies: the roles of green innovation strategic orientation and green intellectual capital," *International Journal of Hospitality Management*, vol. 102, Article ID 103161, 2022.
- [42] K. Haldorai, W. G. Kim, and R. F. Garcia, "Top management green commitment and green intellectual capital as enablers of hotel environmental performance: the mediating role of green human resource management," *Tourism Management*, vol. 88, Article ID 104431, 2022.
- [43] C. Huang and F. Kung, "Environmental consciousness and intellectual capital management," *Management Decision*, vol. 49, no. 9, pp. 1405–1425, 2011.
- [44] B. Marco-Lajara, P. Zaragoza-Saez, J. M. Martínez Falcó, and L. A. Millan-Tudela, "Analysing the relationship between green intellectual capital and the achievement of the sustainable development goals," in *Handbook of Research on Building Inclusive Global Knowledge Societies for Sustainable Development*, pp. 111–129, IGI Global, Hershey, Pennsylvania, 2022.
- [45] Y. S. Chen, "The driver of green innovation and green image - green core competence," *Journal of Business Ethics*, vol. 81, no. 3, pp. 531–543, 2008.
- [46] C. Liu, "Developing green intellectual capital in companies by AHP," in *Proceedings of the En 2010 8th International Conference On Supply Chain Management And Information*, pp. 1–5, IEEE, Hong Kong, China, October 2010.
- [47] M. D. López-Gamero, P. Zaragoza-Sáez, E. Claver-Cortés, and J. F. Molina-Azorín, "Sustainable development and intangibles: building sustainable intellectual capital," *Business Strategy and the Environment*, vol. 20, no. 1, pp. 18–37, 2011.
- [48] C. Chang and Y. Chen, "The determinants of green intellectual capital," *Management Decision*, vol. 50, no. 1, pp. 74–94, 2012.
- [49] C. M. Jardon and A. Dasilva, "Intellectual capital and environmental concern in subsistence small businesses," *Management of Environmental Quality: An International Journal*, vol. 28, no. 2, pp. 214–230, 2017.
- [50] S. Anik and H. Sulistyo, "The role of green intellectual capital and green innovation on competitive advantage of SMEs,"

International Journal of Learning and Intellectual Capital, vol. 1, no. 1, p. 1, 2020.

- [51] Y. S. Chen, S. B. Lai, and C. T. Wen, "The influence of green innovation performance on corporate advantage in Taiwan," *Journal of Business Ethics*, vol. 67, no. 4, pp. 331–339, 2006.
- [52] R. J. Lin, K. H. Tan, and Y. Geng, "Market demand, green product innovation, and firm performance: evidence from Vietnam motorcycle industry," *Journal of Cleaner Production*, vol. 40, pp. 101–107, 2013.
- [53] P. Kivimaa and P. Kautto, "Making or breaking environmental innovation?" *Management Research Review*, vol. 33, no. 4, pp. 289–305, 2010.
- [54] J. Salvadó, G. De Castro, M. Verde, and J. López, Environmental Innovation and Firm Performance: A Natural Resource-Based View, Palgrave Macmillan, London, United Kingdom, 2012.
- [55] G. Albort-Morant, A. Leal-Millán, and G. Cepeda-Carrión, "The antecedents of green innovation performance: a model of learning and capabilities," *Journal of Business Research*, vol. 69, no. 11, pp. 4912–4917, 2016.
- [56] A. Pipatprapa, H. H. Huang, and C. H. Huang, "The role of quality management & innovativeness on green performance," *Corporate Social Responsibility and Environmental Management*, vol. 24, no. 3, pp. 249–260, 2017.
- [57] Y. Zhang, J. Sun, Z. Yang, and Y. Wang, "Critical success factors of green innovation: technology, organization and environment readiness," *Journal of Cleaner Production*, vol. 264, Article ID 121701, 2020.
- [58] A. Mansoor, S. Jahan, and M. Riaz, "Does green intellectual capital spur corporate environmental performance through green workforce?" *Journal of Intellectual Capital*, vol. 22, no. 5, pp. 823–839, 2021.
- [59] S. M. Allameh, "Antecedents and consequences of intellectual capital," *Journal of Intellectual Capital*, vol. 19, no. 5, pp. 858–874, 2018.
- [60] Y. M. Yusoff, M. Nejati, D. M. H. Kee, and A. Amran, "Linking green human resource management practices to environmental performance in hotel industry," *Global Business Review*, vol. 21, no. 3, pp. 663–680, 2020.
- [61] C. H. Wang and W. Juo, "An environmental policy of green intellectual capital: green innovation strategy for performance sustainability," *Business Strategy and the Environment*, vol. 30, no. 7, pp. 3241–3254, 2021.
- [62] C. Pellegrini, F. Rizzi, and M. Frey, "The role of sustainable human resource practices in influencing employee behavior for corporate sustainability," *Business Strategy and the Environment*, vol. 27, no. 8, pp. 1221–1232, 2018.
- [63] J. Xu and Y. Zhang, "Exploring the nonlinear effect of intellectual capital on financial performance: evidence from listed shipping companies in China," *Complexity*, vol. 2021, pp. 1–12, 2021.
- [64] S. A. Yawar and S. Seuring, "Management of social issues in supply chains: a literature review exploring social issues, actions and performance outcomes," *Journal of Business Ethics*, vol. 141, no. 3, pp. 621–643, 2017.
- [65] M. Kohtamäki, J. Vesalainen, S. Henneberg, P. Naudé, and M. J. Ventresca, "Enabling relationship structures and relationship performance improvement: the moderating role of relational capital," *Industrial Marketing Management*, vol. 41, no. 8, pp. 1298–1309, 2012.
- [66] M.-Y. Yusliza, J. Y. Yong, M. I. Tanveer, T. Ramayah, J. Noor Faezah, and Z. Muhammad, "A structural model of the impact of green intellectual capital on sustainable

performance," *Journal of Cleaner Production*, vol. 249, Article ID 119334, 2020.

- [67] V. Matinaro, Y. Liu, T. R. J. S. Lee, J. Poesche, and J. Poesche, "Extracting key factors for sustainable development of enterprises: case study of SMEs in Taiwan," *Journal of Cleaner Production*, vol. 209, pp. 1152–1169, 2019.
- [68] Y. Yu and B. Huo, "The impact of environmental orientation on supplier green management and financial performance: the moderating role of relational capital," *Journal of Cleaner Production*, vol. 211, pp. 628–639, 2019.
- [69] B. Marco-Lajara, P. Zaragoza-Sáez, J. Martínez-Falcó, and E. Sánchez-García, "Green intellectual capital in the Spanish wine industry," in *Innovative Economic, Social, and Environmental Practices for Progressing Future Sustainability*, pp. 102–120, IGI global, Practice, Progress, and Proficiency in Sustainability, Spain, 2022.
- [70] M. Gürlek and M. Tuna, "Reinforcing competitive advantage through green organizational culture and green innovation," *Service Industries Journal*, vol. 38, no. 7-8, pp. 467–491, 2018.
- [71] C. H. Wang, "How organizational green culture influences green performance and competitive advantage," *Journal of Manufacturing Technology Management*, vol. 30, no. 4, pp. 666–683, 2019.
- [72] S. P. Chuang and S. J. Huang, "The effect of environmental corporate social responsibility on environmental performance and business competitiveness: the mediation of green information technology capital," *Journal of Business Ethics*, vol. 150, no. 4, pp. 991–1009, 2018.
- [73] M. Buenechea-Elberdin, J. Sáenz, and A. Kianto, "Knowledge management strategies, intellectual capital, and innovation performance: a comparison between high- and low-tech firms," *Journal of Knowledge Management*, vol. 22, no. 8, pp. 1757–1781, 2018.
- [74] J. Xu, Y. Shang, W. Yu, and F. Liu, "Intellectual capital, technological innovation and firm performance: evidence from China's manufacturing sector," *Sustainability*, vol. 11, no. 19, p. 5328, 2019.
- [75] A. Alrowwad, S. H. AbualoushAbualoush, and R. Masa'deh, "Innovation and intellectual capital as intermediary variables among transformational leadership, transactional leadership, and organizational performance," *The Journal of Management Development*, vol. 39, no. 2, pp. 196–222, 2020.
- [76] S. Liu, Q. Yu, L. Zhang, J. Xu, and Z. Jin, "Does intellectual capital investment improve financial competitiveness and green innovation performance? Evidence from renewable energy companies in China," *Mathematical Problems in Engineering*, vol. 2021, Article ID 9929202, pp. 1–13, 2021.
- [77] H. Ullah, Z. Wang, M. Mohsin, W. Jiang, and H. Abbas, "Multidimensional perspective of green financial innovation between green intellectual capital on sustainable business: the case of Pakistan," *Environmental Science and Pollution Research*, vol. 29, no. 4, pp. 5552–5568, 2022.
- [78] M. Subramaniam and M. A. Youndt, "The influence of intellectual capital on the types of innovative capabilities," *Academy of Management Journal*, vol. 48, no. 3, pp. 450–463, 2005.
- [79] T. Li, L. Liang, and D. Han, "Research on the efficiency of green technology innovation in China's provincial high-end manufacturing industry based on the Raga-PP-SFA model," *Mathematical Problems in Engineering*, vol. 2018, Article ID 9463707, pp. 1–13, 2018.
- [80] M. Ahmad and N. Ahmed, "Testing the relationship between intellectual capital and a firm's performance: an empirical investigation regarding financial industries of Pakistan,"

International Journal of Learning and Intellectual Capital, vol. 13, no. 2/3, p. 250, 2016.

- [81] P. Moran, "Structural vs. relational embeddedness: social capital and managerial performance," *Strategic Management Journal*, vol. 26, no. 12, pp. 1129–1151, 2005.
- [82] J. Huang and Y. Li, "The mediating effect of knowledge management on social interaction and innovation performance," *International Journal of Manpower*, vol. 30, no. 3, pp. 285–301, 2009.
- [83] P. C. ChenChen and S. W. HungHung, "Collaborative green innovation in emerging countries: a social capital perspective," *International Journal of Operations & Production Management*, vol. 34, no. 3, pp. 347–363, 2014.
- [84] D. Ryu, K. H. Baek, and J. Yoon, "Open Innovation with relational capital, technological innovation capital, and international performance in SMEs," *Sustainability*, vol. 13, no. 6, p. 3418, 2021.
- [85] N. Thi Mai Anh, L. Hui, V. D. Khoa, and S. Mehmood, "Relational capital and supply chain collaboration for radical and incremental innovation," *Asia Pacific Journal of Marketing & Logistics*, vol. 31, no. 4, pp. 1076–1094, 2019.
- [86] O. Boiral, "Tacit knowledge and environmental management," *Long Range Planning*, vol. 35, no. 3, pp. 291–317, 2002.
- [87] P. Dickel, J. Hörisch, and T. Ritter, "Networking for the environment: the impact of environmental orientation on start-ups' networking frequency and network size," *Journal* of Cleaner Production, vol. 179, pp. 308–316, 2018.
- [88] L. Cinquini, E. Passetti, A. Tenucci, and M. Frey, "Analyzing intellectual capital information in sustainability reports: some empirical evidence," *Journal of Intellectual Capital*, vol. 13, no. 4, pp. 531–561, 2012.
- [89] N. Aramburu and J. Sáenz, "Structural capital, innovation capability, and size effect: an empirical study," *Journal of Management and Organization*, vol. 17, no. 3, pp. 307–325, 2011.
- [90] I. Maurer, V. Bartsch, and M. Ebers, "The value of intraorganizational social capital: how it fosters knowledge transfer, innovation performance, and growth," *Organization Studies*, vol. 32, no. 2, pp. 157–185, 2011.
- [91] J. J. García-Machado and M. Martínez-Avila, "Environmental performance and green culture: the mediating effect of green innovation. An application to the automotive industry," *Sustainability*, vol. 11, no. 18, p. 4874, 2019.
- [92] H. H. Weng, J. S. Chen, and P. C. Chen, "Effects of green innovation on environmental and corporate performance: a stakeholder perspective," *Sustainability*, vol. 7, no. 5, pp. 4997–5026, 2015.
- [93] D. Kammerer, "The effects of customer benefit and regulation on environmental product innovation," *Ecological Economics*, vol. 68, no. 8-9, pp. 2285–2295, 2009.
- [94] A. Adegbile, D. Sarpong, and D. Meissner, "Strategic foresight for innovation management: a review and research agenda," *International Journal of Innovation and Technology Management*, vol. 14, no. 04, Article ID 1750019, 2017.
- [95] J. Zhang, G. Liang, T. Feng, C. Yuan, and W. Jiang, "Green innovation to respond to environmental regulation: how external knowledge adoption and green absorptive capacity matter?" *Business Strategy and the Environment*, vol. 29, no. 1, pp. 39–53, 2020.
- [96] W. Ben Arfi, L. Hikkerova, and J. M. Sahut, "External knowledge sources, green innovation and performance," *Technological Forecasting and Social Change*, vol. 129, pp. 210–220, 2018.

- [97] K. H. Lee and B. Min, "Green R&D for eco-innovation and its impact on carbon emissions and firm performance," *Journal of Cleaner Production*, vol. 108, pp. 534–542, 2015.
- [98] J. Conding, N. Habidin, A. Zubir, S. Hashim, and N. Jaya, "The structural analysis of green innovation (GI) and green performance (GP) in Malaysian automotive industry," *Research Journal of Finance and Accounting*, vol. 3, no. 6, pp. 172–178, 2012.
- [99] S. Kraus, S. U. Rehman, and F. J. S. García, "Corporate social responsibility and environmental performance: the mediating role of environmental strategy and green innovation," *Technological Forecasting and Social Change*, vol. 160, Article ID 120262, 2020.
- [100] T. Y. Chiou, H. K. Chan, F. Lettice, and S. H. Chung, "The influence of greening the suppliers and green innovation on environmental performance and competitive advantage in Taiwan," *Transportation Research Part E: Logistics and Transportation Review*, vol. 47, no. 6, pp. 822–836, 2011.
- [101] Q. Zhu and J. Sarkis, "Relationships between operational practices and performance among early adopters of green supply chain management practices in Chinese manufacturing enterprises," *Journal of Operations Management*, vol. 22, no. 3, pp. 265–289, 2004.
- [102] N. A. Abu Seman, N. Zakuan, A. Jusoh, M. Arif, and M. Saman, "Green supply chain management: a review and research direction," *International Journal of Managing Value and Supply Chains*, vol. 3, no. 1, pp. 1–18, 2012.
- [103] Y. Huang and Y. Jim Wu, "The effects of organizational factors on green new product success," *Management Decision*, vol. 48, no. 10, pp. 1539–1567, 2010.
- [104] Y. Dong, X. Wang, J. Jin, Y. Qiao, and L. Shi, "Effects of ecoinnovation typology on its performance: empirical evidence from Chinese enterprises," *Journal of Engineering and Technology Management*, vol. 34, pp. 78–98, 2014.
- [105] T. M. Welbourne and M. Pardo-del-Val, "Relational capital: strategic advantage for small and medium-size enterprises (SMEs) through negotiation and collaboration," *Group Decision and Negotiation*, vol. 18, no. 5, pp. 483–497, 2009.
- [106] B. Roxas, M. Battisti, and D. Deakins, "Learning, innovation and firm performance: knowledge management in small firms," *Knowledge Management Research and Practice*, vol. 12, no. 4, pp. 443–453, 2014.
- [107] L. Fleming and O. Sorenson, "Science as a map in technological search," *Strategic Management Journal*, vol. 25, no. 89, pp. 909–928, 2004.
- [108] C. Meyer, B. Skaggs, and M. Youndt, "Developing and deploying organizational capital in services vs. manufacturing," *Journal of Managerial Issues*, vol. 26, no. 4, pp. 326–344, 2014.
- [109] M. Delgado-Verde, J. Amores-Salvadó, G. Martín-de Castro, and J. E. Navas-López, "Green intellectual capital and environmental product innovation: the mediating role of green social capital," *Knowledge Management Research and Practice*, vol. 12, no. 3, pp. 261–275, 2014.
- [110] E. Giuliani, "The selective nature of knowledge networks in clusters: evidence from the wine industry," *Journal of Economic Geography*, vol. 7, no. 2, pp. 139–168, 2007.
- [111] S. Turner, "Networks of learning within the English wine industry," *Journal of Economic Geography*, vol. 10, no. 5, pp. 685–715, 2010.
- [112] L. Cassi, A. Morrison, and A. L. Ter Wal, "The evolution of trade and scientific collaboration networks in the global wine sector: a longitudinal study using network analysis," *Economic Geography*, vol. 88, no. 3, pp. 311–334, 2012.

- [113] G. Cepeda-Carrion, J. G. Cegarra-Navarro, and V. Cillo, "Tips to use partial least squares structural equation modelling (PLS-SEM) in knowledge management," *Journal of Knowledge Management*, vol. 23, no. 1, pp. 67–89, 2019.
- [114] P. Paillé, Y. Chen, O. Boiral, and J. Jin, "The impact of human resource management on environmental performance: an employee-level study," *Journal of Business Ethics*, vol. 121, no. 3, pp. 451–466, 2014.
- [115] W. S. Chow and Y. Chen, "Corporate sustainable development: testing a new scale based on the mainland Chinese context," *Journal of Business Ethics*, vol. 105, no. 4, pp. 519–533, 2012.
- [116] E. Ocde, "Manual de Oslo: Guía para la recogida e interpretación de datos sobreinnovación," 2005, http://www.itq. edu.mx/convocatorias/manualdeoslo.pdf.
- [117] J. Roldan and G. Cepeda, Modelos de Ecuaciones basados en la Varianza: PartialLeast Squares (PLS) para Investigadores en Ciencias Sociales, Universidad de Sevilla, Sevilla, Andalucía, Spain, 2017.
- [118] J. F. Hair, M. Sarstedt, and C. M. Ringle, "Rethinking some of the rethinking of partial least squares," *European Journal of Marketing*, vol. 53, no. 4, pp. 566–584, 2019.
- [119] F. Ali, S. M. Rasoolimanesh, M. Sarstedt, C. M. Ringle, and K. Ryu, "An assessment of the use of partial least squares structural equation modeling (PLS-SEM) in hospitality research," *International Journal of Contemporary Hospitality Management*, vol. 30, no. 1, pp. 514–538, 2018.
- [120] J. Henseler, C. M. Ringle, and M. Sarstedt, "A new criterion for assessing discriminant validity in variance-based structural equation modeling," *Journal of the Academy of Marketing Science*, vol. 43, no. 1, pp. 115–135, 2015.
- [121] W. Reinartz, M. Haenlein, and J. Henseler, "An empirical comparison of the efficacy of covariance-based and variancebased SEM," *International Journal of Research in Marketing*, vol. 26, no. 4, pp. 332–344, 2009.
- [122] J. Henseler, G. Hubona, and P. A. Ray, "Using PLS path modeling in new technology research: updated guidelines," *Industrial Management & Data Systems*, vol. 116, no. 1, pp. 2–20, 2016.
- [123] L. T. Hu and P. M. Bentler, "Fit indices in covariance structure modeling: sensitivity to underparameterized model misspecification," *Psychological Methods*, vol. 3, no. 4, pp. 424–453, 1998.
- [124] B. Byrne, Structural Equation Modeling with EQS: Basic Concepts, Applications, and Programming, Psychology Press, New York, United States, 2008.
- [125] E. G. Carmines and R. A. Zeller, *Reliability and validity* assessment, vol. 17, Sage, Newcastle upon Tyne, United Kingdom, 1979.
- [126] C. Fornell and D. F. Larcker, "Evaluating structural equation models with unobservable variables and measurement error," *Journal of Marketing Research*, vol. 18, no. 1, pp. 39–50, 1981.
- [127] R. Kline, Principles and Practice of Structural Equation Modeling, Guilford Press, New York, N Y, USA, 2011.
- [128] J. Hair, G. Hult, C. Ringle, and M. Sarstedt, A Primer on Partial Least Squares Structural Equation Modeling (PLS-SEM), Sage publications, Newcastle upon Tyne, United Kingdom, 2016.
- [129] A. Gilinsky, S. K. Newton, and R. F Vega, "Sustainability in the global wine industry: concepts and cases," *Agriculture* and Agricultural Science Procedia, vol. 8, pp. 37–49, 2016.
- [130] J. Cohen, Statistical Power Analysis for the Behavioral Sciences, Routledge, New York, NY, USA, 1988.

- [131] C. H. Chang, "The determinants of green product innovation performance," Corporate Social Responsibility and Environmental Management, vol. 23, no. 2, pp. 65–76, 2016.
- [132] M. Hertsgaard, "What climate change means for wine industry," Wired Science, vol. 26, pp. 2011–2023, 2010.
- [133] E. Annunziata, T. Pucci, M. Frey, and L. Zanni, "The role of organizational capabilities in attaining corporate sustainability practices and economic performance: evidence from Italian wine industry," *Journal of Cleaner Production*, vol. 171, pp. 1300–1311, 2018.
- [134] J. Á. del Bríodel Brío, E. Fernández, and B. Junquera, "Management and employee involvement in achieving an environmental action-based competitive advantage: an empirical study," *International Journal of Human Resource Management*, vol. 18, no. 4, pp. 491–522, 2007.
- [135] P. C. Zaragoza-Sáez, E. Claver-Cortés, B. Marco-Lajara, and M. Úbeda-García, "Corporate social responsibility and strategic knowledge management as mediators between sustainable intangible capital and hotel performance," *Journal of Sustainable Tourism*, pp. 1–23, 2020.